Author's response to reviews

Title: Correlates of unequal access to preventive care in China: a multilevel analysis of national data from the 2011 China Health and Nutrition Survey

Authors:

Chi Huang (huangchi_subucm@163.com)
Chao-Jie Liu (c.liu@latrobe.edu.au)
Xiong-Fei Pan (pxiongfei@gmail.com)
Xiang Liu (new9812@126.com)
Ning-Xiu Li (liningxiu@163.com)

Version: 10 Date: 4 May 2016

Author's response to reviews: see over
Dear Editor,

Thank you for your help and suggestions. We have made due changes in the text (MS.8745888981727790) to comply with the requirements from BMC Health Services Research. These changes are highlighted in the text, and are mainly reflected in the Declarations section.

We are looking forward to any comments and would be grateful if the manuscript is handled timely.

Best regards and thank you

Chi Huang

On behalf of Prof Ning-Xiu Li

**Response to Editorial Requests**

1.Methods: Please include the section title 'Methods' as described in our guidelines for authors:

http://bmchealthservres.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article
Re: The section title 'Subjects and methods' was re-described as 'Methods'.

2. Declarations: Please include a complete 'Declarations' section including all the statements described here:

http://bmchealthservres.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article

Re: All the statements described in http://bmchealthservres.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article ('List of abbreviations', 'Ethics approval and consent to participate', 'Consent for publication', 'Availability of data and materials', 'Competing interests', 'Funding', 'Authors' contributions', 'Acknowledgements', and 'Authors’ affiliations') were included in the 'Declarations' section.

3. Availability of data: Please include an 'Availability of data' section as described here:

http://bmchealthservres.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article

Re: 'Availability of data and materials' section was included.
4. List of Abbreviations: Please update the section title 'Abbreviations' to 'List of abbreviations' as described here:

http://bmchealthservres.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article

Re: The section title 'Abbreviations' was updated to 'List of abbreviations'.